Author’s response to reviews

Title: The effect of twin-to-twin delivery time intervals on neonatal outcome for second twins

Authors:

Linnéa Lindroos (linnea.lindroos@vgregion.se)
Anders Elfvin (anders.elfvin@vgregion.se)
Lars Ladfors (lars.ladfors@vgregion.se)
Ulla-Britt Wennerholm (ulla-britt.wennerholm@vgregion.se)

Version: 2 Date: 27 Aug 2017

Author’s response to reviews:

All Editor comments have been addressed.